Author’s response to reviews

Title: Sunflower oil and Mustard oil: Skin barrier function in newborns via a community-based, cluster randomized trial

Authors:

Aimee Summers (aimeesummers@gmail.com)
Marty Visscher (marty.visscher@gmail.com)
Subarna Khatry (skhatry@wlink.com.np)
Jeevan Sherchand (jeevanbsherchand@gmail.com)
Steven LeClerq (sleclerq@mos.com.np)
Joanne Katz (jkatz1@jhu.edu)
James Tielsch (jtielsch@email.gwu.edu)
Luke Mullany (lmullany@jhu.edu)

Version: 2 Date: 29 Nov 2019

Author’s response to reviews:

Response to Editorial Office Review (November 29th, 2019)
BPED-D-19-00538R1; Sunflower oil and Mustard oil: Skin barrier function in newborns via a community-based, cluster randomized trial

Comments
-- Please submit a revised version of your manuscript with the date of registration included in the abstract. The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458, as well as the date of registration. Please note that there should be no space between the letters and numbers of the trial registration number. If registration took place after the first participant was enrolled, please state also “Retrospectively registered” at the end of this section.
Response: We have uploaded a revised version of the manuscript that has the date of trial registration.
-- If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.
Response: We have moved the list of abbreviations to the declarations.
-- Thank you for providing a CONSORT. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.
Response: We have removed the consort diagram from the supplementary files

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response: Our revised manuscript is clean. The tables and figures are uploaded as separate files and are clean versions